Author's response to reviews

Title: Prevalence and correlates of intimate partner violence towards female students of the University of Ibadan, Nigeria Joseph E Umana1*#, Olufunmilayo I Fawole1* and Ikeola A Adeoye1* 1 Department of Epidemiology and Medical Statistics, Faculty of Public Health, College of Medicine, University of Ibadan, Ibadan, Nigeria * Conception and Design: JEU, OIF, Administrative Support: JEU, OIF, IAA, Collection and Assembly of data: JEU, OIF, Data Analysis and Interpretation: JEU, OIF, IAA, Manuscript writing: JEU, OIF, IAA, Final manuscript approval: OIF. All authors read and approved the final manuscript. #Corresponding author JEU, OIF, and IAA are all citizens of Nigeria Email addresses: JEU: umanajoe@yahoo.com OIF: fawoleo@ymail.com IAA: drnichike@yahoo.com Postal address of corresponding author: 20 Akpan Etuk street, Uyo, Akwa Ibom state, Nigeria

Authors:

Joseph E Umana (umanajoe@yahoo.com)
Olufunmilayo I Fawole (fawoleo@ymail.com)
Ikeola A Adeoye (drnichike@yahoo.com)

Version: 5 Date: 14 April 2014

Author's response to reviews: see over
The Executive Editor

BMC series journal

Biomed Central

14/4/14

Response to the concerns of our manuscript’s reviewer - Prevalence and correlates of intimate partner violence towards female students of the University of Ibadan, Nigeria

Formatting changes on our manuscript:

1. A title page has been revised the inclusion of:
   • Country of authors.
   • Postal address of the corresponding author.

Thank you

Joe Umana